# Ontologies in Human-Computer Interaction: A Systematic Literature Review

Simone Dornelas Costa<sup>a,b,\*</sup>, Monalessa Perini Barcellos<sup>b</sup> and Ricardo de Almeida Falbo<sup>b</sup>

 <sup>a</sup> Department of Computer, Federal University of Espírito Santo (UFES), E.S., Brazil E-mail: simone.costa@ufes.br
 <sup>b</sup> NEMO Ontology and Conceptual Modeling Research Group, Department of Computer Science, Federal University of Espírito Santo (UFES), E.S., Brazil

*E-mails: monalessa@inf.ufes.br, falbo@inf.ufes.br* 

**Abstract.** Human-Computer Interaction (HCI) is a multidisciplinary area that involves a diverse body of knowledge and a complex landscape of concepts, which can lead to semantic problems, hampering communication and knowledge transfer. Ontologies have been successfully used to solve semantics and knowledge-related problems in several domains. This paper presents a systematic literature review that investigated the use of ontologies in the HCI domain. The main goal was to find out how HCI ontologies have been used and developed. 35 ontologies were identified. As a result, we noticed that they cover different HCI aspects, such as user interface, interaction phenomenon, pervasive computing, user modeling / profile, HCI design, interaction experience and adaptive interactive system. Although there are overlaps, we did not identify reuse among the 35 analyzed ontologies. The ontologies have been used mainly to support knowledge representation and reasoning. Although ontologies have been used in HCI for more than 25 years, their use became more frequent in the last decade, when ontologies address a higher number of HCI aspects and are represented as both conceptual and computational models. Concerning how ontologies have been developed, we noticed that some good practices of ontology engineering have not been followed. Considering that the quality of an ontology directly influences the quality of the solution built based on it, we believe that there is an opportunity for HCI and ontology engineering professionals to get closer to build better and more effective ontologies, as well as ontology-based solutions.

Keywords: human-computer interaction, ontologies, systematic literature review

# 1. Introduction

Human-Computer Interaction (HCI) is interested in the design, implementation and evaluation of interactive computer systems for human use, along with the phenomena related to this use (Preece et al., 2015; Carroll, 2014). It is a multidisciplinary area that aggregates a vast and multifaceted community, bound by the evolving concept of usability, and the integrating commitment to value human activity and experience as the primary driver in technology (Carroll, 2014). In the last years, the HCI area has expanded to include a myriad of interaction techniques and devices, becoming a host for emerging ubiquitous, handheld and context-aware interactions. Being such a diverse area, it involves a diverse body of knowledge and the landscape of HCI concepts is complex. In addition, as the HCI area continues to mature, new terms are proposed and new meanings are assigned to existing terms, which can result in semantic problems, hamper communication and knowledge transfer (Carroll, 2014; Preece et al., 2015).

1570-5838/0-1896/\$35.00 © 4 - IOS Press and the authors. All rights reserved

<sup>\*</sup>Corresponding author. E-mail: simone.costa@ufes.br.

Ontologies have been successfully used to solve semantics and knowledge-related problems in several domains (e.g., (Ethier et al., 2018; Rodrigues et al., 2020; Zárate et al., 2020; Fonseca et al., 2017)). An ontology is a formal and explicit specification of a shared conceptualization (Gruber, 1995; Studer et al., 1998). It can be used for conceptual description, communication purposes, knowledge management solutions, semantic interoperability, semantic web, reasoning, among others.

Ontologies can be used as conceptual models (reference ontologies) as well as computational artifacts (operational ontologies) that can be read and interpreted by machines (Guizzardi, 2007) allowing automated solutions. The HCI area has benefited from the use of ontologies in many contexts, such as knowledge management in user interface design (Suàrez et al., 2004), accessible web applications engineering (Martín et al., 2010), elicitation of user interface design recommendations (Tourwé et al., 2011), reasoning about the usability of interactive medical devices (Bowen and Hinze, 2012), modeling of user interactions in virtual reality environments (Sokolowski and Walczak, 2018), consistency between user requirements and GUI (graphical user interface) (Silva et al., 2019), among others.

Considering the HCI knowledge diversity and the successful use of ontologies to solve knowledgerelated problems in several domains, we decided to investigate the use of ontologies in the HCI context. Our main goal was to find out (i) how ontologies have been used to support HCI and (ii) how they have been developed. With (i), we intended to understand which portions of the HCI domain have been supported by ontologies and the types of ontology-based solutions that have been adopted to solve HCI problems. With (ii), we aimed to verify the quality of the ontologies used in the HCI domain. This is important because the quality of an ontology directly influences the quality of the solution that uses it. Therefore, we investigated how ontologies have been developed, evaluated and the quality characteristics they exhibit. To achieve our goal, we performed a systematic literature review (SLR).

A SLR aims to evaluate and interpret available research studies relative to a topic, area or phenomenon of interest (Kitchenham and Charters, 2007). By performing a SLR, we want to use a trustworthy and auditable method to make a review about ontologies in HCI and investigate specific aspects about their characteristics and their use in the HCI area. Considering that we also want to provide a panorama of this research topic with some general questions (e.g., about publications year and vehicle) and by means of classification and counting contributions in relation to categories of classification schemas, our SLR also presents characteristics common in systematic mappings (Kitchenham and Charters, 2007; Petersen et al., 2015).

As a result of our study, we found 35 ontologies. These ontologies cover different (and sometimes overlapping) HCI aspects related to user interface (UI), HCI phenomenon, pervasive computing, user modeling/profile, HCI design, interaction experience<sup>1</sup> and adaptive interactive systems. The ontologies have been adopted as conceptual models as well as computational artifacts. They have been used mainly to structure and infer knowledge, support reasoning, serve as a basis for developing systems (e.g., recommender systems) or other approaches (e.g., frameworks) and describe portions of the HCI domain. By analyzing the ontologies, we noticed that many of them do not exhibit important quality characteristics, which can hamper the quality of the produced ontology-based solutions.

The study results contribute to researchers and practitioners interested in HCI and ontologies by providing information about ontologies that have been used in the HCI context and how they have been used. This information allows one to reuse existing ontologies or be inspired by the ontology-based solutions to propose new ones. Moreover, the results reveal how ontologies have been developed, pointing

<sup>&</sup>lt;sup>1</sup>Interaction Experience (IX) is a higher-level concept that encompasses user experience, usability and accessibility (Sauer et al., 2020)

out ontology engineering practices that should be considered when developing HCI ontologies, which can indicate opportunities for collaboration between HCI and ontology engineer professionals. Furthermore, gaps that arose from the study results can be addressed in new research, providing advances in the state of the art.

This paper is organized as follows. Section 2 presents the background for this paper and discusses some related works. Section 3 presents the followed research protocol. Section 4 addresses data extraction and synthesis. Section 5 discusses the findings. Section 6 regards the study limitations. Finally, Section 7 concludes the paper.

# 2. Background

In this section, we briefly present the background for the paper. Section 2.1 concerns HCI and ontologies and Section 2.2 discusses some related works.

# 2.1. HCI and Ontologies

HCI interest in interactive systems and their impact on people's life has promoted the study and practice of usability, a key aspect related to user efficiency and satisfaction when interacting with the computer. HCI is also concerned with other important qualities characteristics such as user experience, accessibility and communicability. It is a multidisciplinary area, involving knowledge from multiple fields, such as ergonomics, cognitive science, user experience, human factors, among others (Carroll, 2014; Preece et al., 2015; Sutcliffe, 2014). Since HCI is a complex knowledge area, ontologies can be useful to capture, organize and share the involved knowledge.

According to Scherp et al. (2011), ontologies can be organized in a three-layered architecture that discriminates between foundational, core and domain ontologies. *Foundational ontologies* aim at modeling the very basic and general concepts and relations that make up the world (e.g., objects, events, participation, parthood). *Core ontologies* provide a refinement to foundational ontologies by adding detailed concepts and relations in a specific field (such as service, process, organizational structure) that still spans across various domains. Finally, *domain ontologies* describe knowledge that is specific for a particular domain, such as a domain-specific medical ontology describing the anatomy of the human body. Its concepts and relations can make use of or be based on foundational ontologies or core ontologies by specializing their concepts.

In another classification, orthogonal to the one proposed by Scherp et al. (2011), Guizzardi (2007) differentiates between reference and operational ontologies. A *reference ontology* is a conceptual model, whose aim is to make a clear and precise description of the elements (concepts, relations, properties) of a conceptualization for the purposes of communication, learning and problem-solving. A reference ontology is to be used in an offline manner to assist humans in tasks such as meaning negotiation and consensus establishment. Such a consensual conceptual model can be coded in a computational language (e.g., OWL) to be then deployed in a computational environment. This implementation version is an *operational ontology*. Unlike reference ontologies, operational ontologies are not focused on representation adequacy, but are designed with the focus on guaranteeing desirable computational properties. In the HCI domain, ontologies have been used for several purposes. Some works (such as the ones by Storrs (1994), and Mezhoudi and Vanderdonckt (2015)) have used ontologies as a conceptual model to conceptualize the HCI domain. Others have used only operational ontologies. For example, in the work by Garay-Vitoria et al. (2019), an operational ontology about affective computing was used as a basis for developing Affective Interactive Systems. There are also works that take advantage of both reference and operational ontologies. Bakaev et al. (2019), for example, used a reference ontology about UI and UI measurement to structure and organize knowledge, while an operational version of the ontology was used in a tool to assess web UI using different metrics.

# 2.2. Related Work

In the literature, some works describe secondary studies that investigate aspects related to HCI. For example, Guerino and Valentim (2020) conducted a systematic mapping study to investigate usability and user experience evaluation technologies used by researchers and developers in software with natural user interfaces. Li et al. (2020), in turn, carried out a systematic literature review and investigated the essence, process and result of how tangible interaction with light is conceptualized. Other examples of systematic literature review in the HCI domain are the work by Maalej and Kallel (2020), which reports the effectiveness of using keystroke dynamics biometrics in recognizing emotions, and the paper by Gubert et al. (2020), which addresses the use of context-awareness in health, defining a taxonomy and identifying challenges and open questions.

We did not find a secondary study that focuses on ontologies in HCI. Hence, in the following paragraphs, we summarize some secondary studies that provide reviews of the use of ontologies in embedded systems (Sousa et al., 2016), software process assessment (Tarhan and Giray, 2017), higher education (Tapia-Leon et al., 2018) and liver diseases (Messaoudi et al., 2020).

Sousa et al. (2016) report a systematic mapping study performed to collect evidence about the need for ontologies in embedded systems development and the benefits of using ontologies in that context. The authors investigated the software processes in which ontologies were applied, the used ontology languages, the purpose of using ontologies, the obtained benefits, among others. 19 publications were selected for data extraction. Considering the purpose of use, ontologies were applied mainly to domain knowledge representation and to provide a common vocabulary to communication between the engineers and requirements representation. The benefits of using ontology in embedded systems were to guide requirements elicitation, deal with ambiguity and inconsistencies between the requirements, facilitate communication between engineers and promote non-functional requirements standardization.

Tarhan and Giray (2017) performed a systematic literature review to investigate the state of the art of the use of ontologies in software process assessment. Similar to the work by Sousa et al. (2016), the study investigated the software processes where ontologies were applied, the used ontology languages, the purpose of using ontologies, among others. Data was extracted from 14 publications.  $OWL^2$  was the predominant representation language and ontologies were used mainly in general software process (i.e., there is no indication of any specific process), whereas some of the ontologies were applied in testing, project management and quality processes. Regarding the purpose of ontology use, most of the studies focused on knowledge representation, followed by ontology mapping and knowledge reasoning over their represented knowledge.

Tapia-Leon et al. (2018) carried out a systematic mapping focusing on uses of ontologies in higher education, as well as the methods, tools, languages and vocabularies used in the ontology development process. 52 publications were selected for data extraction. The results showed that ontologies were used mainly in higher education enterprise (i.e., to represent knowledge regarding components of a higher education institution), e-learning and academic recommendation. Concerning ontology development,

<sup>&</sup>lt;sup>2</sup>World Wide Web Consortium (W3C) Web Ontology Language, https://www.w3.org/OWL/

the authors said that most of the papers (81%) do not specify the method used to develop the ontology. Moreover, most of the times, it was not possible to identify reused ontologies. Protegé<sup>3</sup> was the most cited tool, but most of the papers do not specify the used tool. Domain ontologies are the most common type of ontology, being represented by computable models in most of the analyzed works. As such, OWL is the primary used language, followed by RDF<sup>4</sup>/RDFS<sup>5</sup>. Most proposals define a theoretical or conceptual framework for the application or use of ontologies. Only 21% of the papers presented a partial implementation or testing in a specific use scenario.

Finally, Messaoudi et al. (2020) report a systematic literature review focusing on the use of ontologies to represent liver diseases. As in the aforementioned studies, the authors investigated the purpose of using ontologies in this context. The authors also looked for concepts with higher frequency rates. As a result, after extracting data from 39 publications, the authors noticed that the majority of the found ontologies were computational models. Therefore, the most recurrent uses were to process large medical data volumes, support decision system in some cases and also diagnosis process acceleration and assistance. Concerning the concepts, "TypesLiverDisease", "RiskFactors", and "Symptoms" had the highest frequency rate and, thus, the authors considered them the most useful ones.

Our study presents some similarities and also differences when compared to the ones cited in this section. As for similarities, as the works by Gubert et al. (2020), Guerino and Valentim (2020), Li et al. (2020), and Maalej and Kallel (2020), our work is a secondary study that investigates aspects in the HCI domain. Moreover, like in the works by Messaoudi et al. (2020), Sousa et al. (2016), Tapia-Leon et al. (2018), and Tarhan and Giray (2017), our work regards the use of ontologies in a particular domain. However, there are important differences between our work and the others. Although the studies carried out by Gubert et al. (2020), Guerino and Valentim (2020), Li et al. (2020), and Maalej and Kallel (2020) are secondary studies investigating HCI aspects, different from our work, they are not concerned with the use of ontologies in HCI. On the other side, even though the works by Messaoudi et al. (2020), Sousa et al. (2016), Tapia-Leon et al. (2018), and Tarhan and Giray (2017) concern the use of ontologies, they are not focused on the HCI domain. Nevertheless, since these studies investigated the use of ontologies in a particular domain, some investigated questions are similar to the ones investigated in our study. For example, all the studies investigated the purpose of using ontologies. As in the works by Tapia-Leon et al. (2018) and Sousa et al. (2016), we also investigated which portions of the domain of interest have been addressed by ontologies. Like Tapia-Leon et al. (2018), we also investigated aspects related to ontology engineering aiming to verify the quality of the proposed ontologies. However, we used a set of quality characteristics different from that work. Some results obtained in these studies can also be compared to ours. For example, similar to Tapia-Leon et al. (2018), we also noticed that many publications are not clear about the method used to develop the proposed ontologies. Considering domain coverage, like Sousa et al. (2016), we noted that an ontology can address different portions of the domain of interest. As for ontology use, aligned to the works by Sousa et al. (2016) and Tapia-Leon et al. (2018), our study showed that ontologies have been used mainly for knowledge management, which includes knowledge representation. Moreover, OWL has also been the main language to represent ontologies as computational artifacts in the HCI domain.

<sup>&</sup>lt;sup>3</sup>Open-source ontology editor, https://protege.stanford.edu

<sup>&</sup>lt;sup>4</sup>Resource Description Framework, https://www.w3.org/RDF/

<sup>&</sup>lt;sup>5</sup>RDF Schema, https://www.w3.org/wiki/RDFS

# 3. Research Protocol

To perform our study, we followed the process defined by Kitchenham and Charters (2007), which comprises: planning, when the research protocol is defined with the purpose of supporting study replicability as well as helping researchers to avoid bias when conducting the study; conducting, when the protocol is executed and data are extracted, analyzed and recorded; and reporting, when the results are recorded and made available to potential interested parties. In this section, we present the main parts of the research protocol used in the study.

The *study goal* was to understand how ontologies have been used in the HCI domain and how they have been developed. In our study, we refer to these ontologies as HCI ontologies. By understanding how ontologies have been used in the HCI domain, it is possible to verify if they have been used like in other domains (e.g., for reasoning, classification), if new applications have been explored and if there are opportunities for the HCI area to benefit from ontologies use. For example, by identifying if there is an HCI sub-domain favorable to the use of ontologies and that has not explored ontologies use yet, one can be inspired to propose ontology-based solutions to aid HCI practices in that sub-domain. By investigating how HCI ontologies have been developed, it is possible to verify if there has been concern with a ontology quality issues and if there is room for closer interaction between HCI and ontology engineer professionals. For achieving the study goal, we defined the *research questions* presented in Table 1.

RQ	Research Question	Rationale						
RQ1	When and in which type of vehicle have the studies been published?	Provide an understanding on when and where (jour- nal/conference/workshop) publications addressing HCI ontologies have been published, to analyze the maturity of the research topic.						
RQ2	Which aspects of the HCI domain have been covered by the existing ontologies?	Identify the portions of the HCI domain covered by the ontologies to analyze the comprehensiveness of the HCI ontologies and of their use in HCI domain.						
RQ3	What have been the uses of HCI ontolo- gies?	Identify the types of solutions that have been supported by ontologies and understand the purposes of using ontologies in the HCI domain.						
RQ4	Have the HCI ontologies been devel- oped by following ontology engineer- ing methods?	Understand how HCI ontologies have been developed, aiming to analyze if good ontology engineering practices have been adopted.						
RQ5	How have the HCI ontologies been evaluated?	Verify if (and how) HCI ontologies have been evaluated to analyze if ontology evaluation has been a concern in HCI ontologies.						
RQ6	Which quality characteristics have HCI ontologies exhibited?	Identify the HCI ontologies quality characteristics to verify if the on- tologies used in the HCI domain present good quality level.						

 Table 1

 Research questions and their rationale

The *search string* adopted in the study contains two groups of terms joined with the operator AND. The first group includes terms related to HCI. The second group includes terms related to Ontology. Within the groups, we used OR operator to allow synonyms. The following search string was used: (*"user interface ontology" OR "user interface ontologies" OR "UI ontology" OR "UI ontologies"*) OR ((*"human-computer interaction" OR "HCI" OR "user interface design" OR "user interface sign" OR "user interface design" OR "user interface design" OR "user interface design" OR "user interface assessment" OR "user interface assessment" OR "user interaction") AND (<i>"ontologies"*)

*OR* "*ontology*")). For defining the search string, we followed the strategy of iteratively improving the search string until obtaining satisfactory results in terms of the number of publications and their relevance (Petersen et al., 2015). We started with a very general search string including terms such as "human-computer interaction", "HCI", "user interface" and "ontology". After running some tests, we noticed that the term "user interface" caused many irrelevant papers to be returned. Thus, we refined the search string by combining the term "user interface" (and synonyms) with other terms such as "ontology", "design", "evaluation" and "assessment".

The *sources* used in the study were digital libraries that satisfy the following criteria (Costa and Murta, 2013): (i) it allows using logical expressions or a similar mechanism for performing searches; (ii) its search mechanism supports full-length searches; (iii) it is available in the researchers' institution; and (iv) it covers the research area of interest, namely Computer Science. The following sources were selected: Scopus and Engineering Village. Scopus is one of the largest databases of peer-reviewed literature. It indexes papers from other important sources such as IEEE, ACM and Science Direct, providing useful tools to search, analyze and manage scientific research. Engineering Village is also a citation indexing platform, which indexes complementary sources. Initially, we have also selected the Web of Science database. However, it did not return any additional paper when we searched publications until 2018. Thus, we decided to exclude it from the set of sources.

**Publications selection** was performed in four steps. In *Preliminary Selection and Cataloging (S1)*, the search string was applied in the search mechanism of each digital library used as source of publications (we limited the search scope to title, abstract and keywords metadata fields). After that, we performed the *Duplications Removal (S2)*, i.e., publications indexed in more than one digital library were identified and duplications were removed. In *Selection of Relevant Publications – 1st filter (S3)*, the abstracts of the selected publications were analyzed considering the following inclusion (IC) and exclusion (EC) criteria: (IC1) the paper presents an ontology about the HCI domain; (EC1) the paper does not present an ontology about the HCI domain; (EC2) the paper does not have an abstract; (EC3) the paper was published only as an abstract; (EC4) the paper is a secondary study, tertiary study, editorial, summary of keynote, tutorial or the proceedings of a scientific event; (EC5) the paper is not written in English. Finally, in *Selection of Relevant Publications – 2nd filter (S4)*, the full text of the publications selected in S3 was read and analyzed considering the aforementioned criteria plus the following: (EC6) the paper is an older version (outdated) of another paper already considered; and (EC7) we did not have access to the full text of the paper. The (EC6) served to avoid repetition and (EC7) to treat publications whose full text was not available for the researchers.

We used the StArt tool<sup>6</sup> to support publications selection. To consolidate data, publications returned in the publication selection steps were cataloged and stored in spreadsheets. We defined an id for each publication and recorded the publication title, authors, year, and vehicle of publication. Data from publications returned in S4 were extracted and organized into a data extraction table-oriented to the research questions. The spreadsheets produced during the study can be found in our SLR package (Costa et al., 2020).

The first author performed publications selection and data extraction. The second and third authors reviewed both. Once data has been validated, the first author carried out data interpretation and analysis, and again second and third authors reviewed the results. Discordances were discussed and resolved. Quantitative data were tabulated and used in graphs and statistical analysis. Finally, the three authors

<sup>&</sup>lt;sup>6</sup>http://lapes.dc.ufscar.br/tools/start\_tool

performed qualitative analysis considering the findings, their relation to the research questions and the study purpose.

# 4. Data Extraction and Synthesis

428

The study considered papers published until 2019. Searches were conducted for the last time in January 2020. Figure 1 illustrates the followed process and the number of publications selected in each step. To search the digital libraries, the search string went through syntactic adaptations according to particularities of each source.



Figure 1. Search and Selection Process.

In the 1st step, as a result of searching the selected sources, a total of 1906 publications was returned. In the 2nd step, we eliminated duplicates, achieving 899 publications (reduction of approximately 53%). In the 3rd step, we applied the selection criteria over the abstract, resulting in 117 papers (reduction of approximately 94%). At this step, we only excluded publications that were clearly unrelated to the subject of interest. In case of doubt, the paper was taken to the next step. Finally, in the 4th step, the selection criteria were applied considering the full text, resulting in 38 publications (reduction of approximately 68%). It is important to highlight that the high number of excluded publications in the 3rd and 4th steps is because we decided to use a more comprehensive search string, which brings a higher number of publications that do not meet the inclusion criteria, but reduces the risk of not selecting relevant publications.

From the 38 selected publications, we identified 35 different ontologies because some publications present complementary parts of the same ontology. Table 2 shows the list of identified ontologies and corresponding publications. Along this and the next sections we refer to the ontologies by using the id listed in the table. Detailed information about the selected publications, including a brief description and extracted data, can be found in our SLR package (Costa et al., 2020). After Table 2, we present the data synthesis for each research question.

ID	Ontology	Brief description and some concepts	Ref.
#01	Ontology of Multi- party Interaction	It is a conceptual model, based on references such as (Norman, 1986), which addresses the HCI phenomenon and involved parts (i.e., user and system). Includes concepts such as Participants, User and Interaction.	(Storrs, 1994)
#02	UI Design Process Ontology	It is a conceptual model on task modeling and interaction with a focus on UI design. Some concepts: Task, Interaction, Action.	(Suàrez et al., 2004)
#03	Impairment-User In- terface Ontology	It is a conceptual model on UI and adaptation for user impairment. Ad- dresses concepts such as Impairment, User, InterfaceAdaptation.	(Karim and Tjoa, 2006)

ID	Ontology	Brief description and some concepts	Ref.
#04	DREAMS Ontology	It presents a conceptual model and a small extract of the operational ontology implemented in OWL and RDF that addresses user stereotypes and related aspects for interaction and devices capabilities in massive networking scenarios. Includes concepts such as Social_Role, Device, Input modality.	(Korfiatis and Constantiou, 2006)
#05	SOUPA Extension	It is a conceptual model which addresses multimodal pervasive systems, devices and environments in multimodal interaction. Some concepts: I/O Device, Sensor, Service.	(Coronato and De Pietro, 2007)
#06	HCI Frame-based On- tology	It is a conceptual model on task modeling and interaction with a focus on UI design regarding guidelines structures. Includes concepts such as Target User, Design Property, Usability Requirement.	(Bakaev and Avdeenko, 2010, 2012)
#07	Ontology for Interac- tive Adaptive Systems	It is an operational ontology, implemented in OWL, for adaptive interac- tive systems and its facets such as user, system, user-system interaction and adaptation. Includes concepts such as DialogSystem, Interaction, Adaptation, UserModeIItem.	(Bezold and Minker, 2010)
#08	Interaction Ontology	It presents a small extract of an operational ontology implemented in OWL that addresses interaction patterns and web UI regarding struc- tural and visual aspects. Some concepts: Widget, Event, Presentation Properties.	(Celino and Corcoglioniti, 2010)
#09	Abstract Widget On- tology	It is a conceptual model, addressing web UI elements and concepts such as Abstract Interface Element, Element Exhibitor, Variable Capture.	(Martín et al., 2010)
#10	User Interface Ontol- ogy	It is an operational ontology implemented using OWL/RDF. It ad- dresses interface entities and relationships, including concepts such as Widget, Textual, Frame.	(Shahzad et al., 2011)
#11	Core HCI Ontology	It is a conceptual model of UI design addressing input and output modalities to be chosen according to the context of use. Some concepts: Location Modality Device	(Tourwé et al., 2011)
#12	The name is not cited in the paper	It presents a conceptual model and an operational version that addresses interactive modal devices and situations of use. Includes concepts such as Device, DeviceRestrictedByLocation, Situation.	(Bowen and Hinze, 2012)
#13	Gesture Ontology	It is a conceptual model that addresses gesture interaction and aspects involved during the process of producing a gesture. Includes concepts such as User, Body Movements, Function-oriented Gesture.	(Chera et al., 2012)
#14	User Interaction Con- text Ontology	It is a conceptual model about user interaction and interaction context addressing the characterization of user actions and resources used dur- ing his/her interaction. It has an operational version implemented using OWL and includes concepts such as Action, Resource, Task, User.	(Devaurs et al., 2012)
#15	RaUL Ontology	It is a conceptual model, a markup ontology, about web UI elements related to the structure of a web form. It has an operational version implemented using RDF and includes concepts such as Page, Widget Element, Button.	(Haller et al., 2012)
#16	PersonasOnto	It is a conceptual model about persona characterization and personas aspects related to usability test. It has an operational version implemented using OWL. Some concepts: Person. Persona. Usability Test.	(Negru and Buraga, 2012, 2013)
#17	An ontology for Per- sonas	It is a conceptual model about persona characterization. A small extract of an operational ontology implemented in OWL/RDF is presented. In- cludes concepts such as Person, Persona, Context, Goal.	(Salma et al., 2012; Salma and Marouane, 2016)
#18	Haptic Applications Software Modeling Ontology	It is an operational ontology, implemented using OWL/SWRL about haptic HCI and pathways that haptic information follows between the human and the machine haptic system. Some concepts: Haptic-Devices, Computer haptics, Haptic-Feedback.	(Myrgioti et al., 2013)
#19	End-to-End Interac- tion Ontology	It is a conceptual model and an operational ontology (implemented in OWL) that addresses elements of interaction under the WIMP (Win- dows, Icons, Menus, Pointer) UI perspective. Includes concepts such as Interaction, Interface, User.	(Zamzami et al., 2013)

ID	Ontology	Brief description and some concepts	Ref.
#20	Ergonomics of Human-Computer Interface Ontology	It is a conceptual model concerning ergonomics of UI and aspects such as recommendations, graphic design and evaluation. Includes concepts such as HCI Ergonomics, Interface_Evaluation, Design_And_Evalua- tion.	(Elyusufi et al., 2014)
#21	Interaction Event On- tology	It is a conceptual model about interaction event in social media, cover- ing aspects related to users' (social) interactions. Some concepts: Event, InteractionEvent, SocialRole.	(Kabir et al., 2014)
#22	User's Feedback On- tology	It is a conceptual model that addresses user interaction feedback con- cerning aspects related to structure, credibility and information value. Includes concepts such as Role, Significance, Modality.	(Mezhoudi and Vanderdonckt, 2015)
#23	Web Design Domain Ontology	It is a conceptual model about web UI design property and web UI structure and elements. Some concepts: Interface element, menu, design property, Interface design property.	(Bakaev and Gaedke, 2016)
#24	Ontological Frame- work for Universal Acess to ICT	It is a conceptual model about universal accessibility in information and communication technologies according to the user needs and pref- erences. Includes concepts such as UserInterface, UserCharacteristic, UserCondition.	(Koutkias et al., 2016)
#25	Interaction Ontology	It is a conceptual model about craftswoman capacities and appropri- ated input/output (I/O) interaction modalities, modes and mediums. It is implemented using OWL/SWRL and includes concepts such as Craftswoman, Motor Capacity, Modality, Medium,	(Lebib et al., 2016)
#26	Ontological Model of Adaptive User Inter- face for People with Disabilities	It is a conceptual model about adaptive UI for people with disabilities, covering UI elements and user characterization. An operational version is implemented using Java and Jena. Includes concepts such as User, Disability. Interface Element Type.	(Kultsova et al., 2017)
#27	Usability Guideline Ontology	It is a conceptual model about usability guidelines description and structure. It is implemented using OWL and includes concepts such as Guideline, Guideline Element, Content Type.	(Robal et al., 2017)
#28	PG-Ontology	It is a conceptual model about pervasive game, concerning its descrip- tion, elements/components and their relationships from the user expe- rience perspective. Includes concepts such as Pervasive_game, Player, Game.	(Arango-López et al., 2018)
#29	Context Model Centre Ontology	It is a conceptual model about context-awareness, addressing context as the information that is used to characterize the relevant objects and entities that affect computing tasks. Some concepts: Context Domain, Environment Context, User Context.	(Meng, 2018)
#30	Accesibilitic Ontol- ogy	It is a conceptual model about characterization of user needs and ca- pabilities, assistive devices/software tools used to support accessibility and the user's interaction taking into account capabilities. An opera- tional version is implemented using OWL. Includes concepts such as User, Capacity, ActivityParticipation, SupportAssistance.	(Romero Mar- iño et al., 2018)
#31	Ontologies of Interac- tion Channels – Vir- tual Reality System and Environment	It is a conceptual model about characteristics of interaction channels of Virtual Reality system and environment. Includes concepts such as InteractionChannel, Forms_of_interaction, Spacial_navigation.	(Sokolowski and Walczak, 2018)
#32	Web UI Measurement	It is a conceptual model and an operational ontology (implemented in OWL) about the organization of UI metrics and services used in the UI measurement. Includes concepts such as Service, Metric, Attribute, Interface	(Bakaev et al., 2019)
#33	Affinto Ontology	It is a conceptual model and an operational ontology (implemented in OWL) about affective HCI, which addresses affective states and different factors of its context of use. Some concepts: Environmental_property, System_property, AffectiveP_properties.	(Garay-Vitoria et al., 2019)

ID	Ontology	Brief description and some concepts					
#34	IoRT Ontology	It is a conceptual model about the user characteristics and interaction in IoRT (Internet of Robotic Things), addressing relevant aspects for personalized, context-aware task generation. An operational version is implemented using OWL. Includes concepts such as Preferred Interac- tion. PreferredRobot Interactions, Task, Profile.	(Mahieu et al., 2019)				
#35	The name is not cited in the paper	It is a conceptual model and an operational ontology (implemented in OWL) about the presentation and behavior of interactive components (graphical UI) used in web and mobile applications. It also addresses the structure of user stories, tasks, scenarios and prototypes. Some concepts: Event, Action, Condition, Menu, clickOn.	(Silva et al., 2019)				

**Publication year and vehicle (RQ1)**: Figure 2 shows the distribution of the 38 selected publications over the years and the distribution of publications considering the vehicle of publication. The oldest paper addressing ontologies in the HCI context we found in the study was published in 1994. Since then, papers addressing this subject have been published in journals and conferences (no workshop publication was found). Conferences have been the main forum, encompassing 68.4% of the publications (26 out of 38). 12 papers (31.6%) were published in journals. In Figure 2 we omitted years in which no publication was found. Hence, looking at the x-axis it is possible to notice some intervals where no publication was identified. The more expressive is the 9-years gap between 1994 and 2004.





**Coverage of the HCI domain (RQ2)**: Figure 3 shows the HCI aspects addressed by the ontologies identified in the study. The categorization used to classify the aspects was established based on the extracted data, HCI literature and also in topics of interest cited in important HCI conferences<sup>7,8</sup>. First, we analyzed each ontology to identify the aspects it addresses, according to the sub-domain (or sub-domain portion) covered by the ontology. In this first moment, we kept the terminology used in each publication. For example, ontology #13 addresses the HCI phenomenon by means of gesture, while #07 covers aspects related to adaptive interactive systems and user modeling. After that, we grouped similar aspects. For example, all ontologies addressing the HCI phenomenon, regardless of the mode of interaction, were

<sup>&</sup>lt;sup>7</sup>INTERACT 2021, https://www.interact2021.org/

<sup>&</sup>lt;sup>8</sup>CHI 2021, https://chi2021.acm.org/

432

grouped in HCI phenomenon. As a result, we reached 13 groups. Finally, we analyzed them and, considering HCI sub-domains found in the literature and topics of HCI conferences, we defined seven more general categories, namely: *User Interface* (addresses user interface types and elements, such as widget, graphical user interface); *HCI phenomenon* (deals with the interaction phenomenon regardless of the mode of interaction); *Pervasive Computing* (involves ubiquitous, context-aware, mobile and pervasive computing in the context of interactive computer systems) (Satyanarayanan, 2001; Surve and Ghorpade, 2017); *User Modeling / Profile* (addresses user characterization such as impairment, social roles, user modeling, persona, needs and preferences, capacity, disability) (Schiaffino and Amandi, 2009; Fischer, 2001); *HCI Design* (refers to aspects related to processes, methods and principles of the HCI design such as persona (as a method), usability testing, HCI evaluation and user interface measurement); *Interaction Experience* (addresses interactive computer system quality characteristics such as usability, user experience and accessibility) (Sauer et al., 2020); and *Adaptive Interactive System* (concerns adaptive and adaptable interactive computer systems). Each of the 35 ontologies was, thus, categorized according to the seven categories. For example: #07 was classified in the *Adaptive Interactive System* and *User Modeling / Profile* categories.



Figure 3. HCI aspects covered by the HCI ontologies.

*User Interface* (UI) has been the HCI aspect most addressed by the HCI ontologies, being covered by 12 of them (34.29%). In this context, some ontologies focus on the UI elements (#08, #09, #10, #15, #19, #23, #26, #35), addressing them from graphical (#10, #35), web (#08, #09, #15, #23, #26) and WIMP (#19) perspectives. Other ontologies address other aspects of UI, such as UI adaption (#03), interface modality (#04, #25) and UI measurement (#32). *HCI phenomenon* has been the second HCI aspect most addressed, being treated in 11 ontologies (31.43%). Some ontologies deal with specific modes of interactions such as gesture (#13), haptic (#18), virtual reality (#31), affective (#33) and IoRT (#34). #25, in turn, addresses several modes of interaction, representing them in categories such as direct manipulation, language, visual, hearing and tactile. #01 describes HCI as multiparty, concerning one or more participants. #14, #19, #21 and #35 represent HCI by means of events, actions and tasks.

*Pervasive Computing, User Modeling / Profile* and *HCI Design* were addressed by 10 (28.57%) ontologies. In *Pervasive Computing*, six ontologies treat context-awareness by means of interactive medical devices in multiple contexts and situations of use (#12), user interaction context for automatic task detection (#14), context-awareness in mobile adaptation (#29), context-based approach in virtual reality (#31),

context of use factors that influence affective states (#33) and context-aware systems for personalized interaction tasks in IoRT (#34). Three ontologies treat mobile computing dealing with mobile devices capacities (#04), accessibility (#24) and performance (#29). Two ontologies treat pervasive systems by means of pervasive game description (#28) and proactive environments (#05).

In *User Modeling / Profile*, ontologies address user impairment (#03), social roles (#04), user modeling (#07), persona (#16, #17), user capacity (#25, #30), disability (#26), and needs and preferences (#24, #34). Concerning *HCI Design*, #13 focused on HCI design for gesture interaction, #02, #06, #08, #11, #16, #17, #20, #23 on user interface design. #16 encompasses usability testing under the persona perspective, #20 concerns user interface ergonomic guidelines and #32 focused on user interface metrics and measurement.

Eight (22.86%) ontologies address *Interaction Experience*, dealing with UI accessibility (#03, #25, #26) and user needs targeting accessibility issues (#24, #30). #20 addresses ergonomics by means of recommendations, #27 describes usability guidelines while #28 deals with user experience in the context of games. Three (8.57%) ontologies address *Adaptive Interactive System*, dealing with adaptive interactive system as a whole (#07), adaptive interactive system based on user feedback (#22) and UI adaptation for user impairment (#03).

By combining when the studies have been published (RQ1) and HCI aspects covered by the ontologies (RQ2) it is possible to identify changes of focus in the HCI domain regarding the use of ontology over the years. Figure 4 illustrates this information and presents trend lines showing increasing in some HCI aspects addressed by ontologies over the years. In order to not visually pollute the graph, in the figure we show only aspects that have been target of increasing interest. As in Figure 2, in Figure 4 we omitted years in which no publication was found. Hence, the figure considers only years where we identified publications about the investigated topic.



Figure 4. HCI aspects covered by the HCI ontologies over the years.

As showed in Figure 3, *User Interface* and *HCI phenomenon* have been the HCI aspects addressed in more ontologies. When looking at Figure 4, it is possible to notice an increasing tendency to address the

*HCI phenomenon.* On the other side, although *User Interface* has been target of many ontologies, when looking at the distribution of them over the years, it is possible to notice a stable trend in the number of ontologies addressing this aspect (trendline not showed in the graph), while *Pervasive Computing* and *Interaction Experience* have an increasing trendline.

In order to perceive information that may not be evident only by looking at Figure 4, we grouped data presented in Figure 4 by period of time. Thus, we grouped publications in periods of five and four years: 1994-2010 (1994, 2004, 2006, 2007 and 2010); 2011-2015 (from 2011 to 2015); and 2016-2019 (from 2016 to 2019).

In the 1994-2010 period there are nine (23.68%) papers (#01, #02, #03, #04, #05, #06 (2010), #07, #08, #09) with an average of 1.8 articles published by year. In this interval, *User Interface* stands out, being treated by four (44.44%) ontologies (#03, #04, #08 and #09). There is an average of 2.29 HCI aspects addressed per publication, showing that a single ontology often addresses more than one HCI aspect (as Figure 3 shows). Moreover, there is an average of three aspects addressed per year. For instance, ontology #03 was developed to help customize/adapt the UI according to user impairments, aiming to improve user interaction for people with special needs. Thus, it addresses UI (*User Interface*), UI adaptation (*Adaptive Interactive System*), user impairment (*User Modeling / Profile*) and accessibility (*Interaction Experience*). #09, in turn, addresses web user interfaces and web user interface design (i.e., *User Interface* and *HCI Design*), being used in an aspect-oriented approach for engineering web applications.

In the 2011-2015 period there was an increase of 66.67% of publications when compared to the previous period. There were 15 publications (39.47%) with an average of 3.0 publications per year, four HCI aspects per year, and an average of 3.14 HCI aspects addressed per publication. *HCI Design* was the aspect most treated by ontologies in this period, being covered by seven (46.67%) publications (#06, #11, #13, #16, #17, #20). Four of them (#06, #13, #16, #17) contributed to the high number of publications in 2012 (as shown in figures 2 and 4).

In the 2016-2019 quadrennium, 14 (36.84%) papers were published, with an average of 3.5 publications and 8.0 aspects per year and 3.43 HCI aspects per publication. Three aspects stand out equally (each one was treated by six (42.86%) ontologies): *Pervasive Computing* (#24, #28, #29, #31, #33, #34), *User Modeling / Profile* (#17, #24, #25, #26, #30, #34) and *Interaction Experience* (#24, #25, #26, #27, #28, #30).

Use of Ontologies in HCI (RQ3): When investigating how ontologies have been used in the HCI domain, we first looked at the types of ontology-based solutions that have been used and, thus, we investigated their use in the HCI domain. Figure 5 shows the identified types of ontology-based solutions.

The most common ontology-based solution has been *Knowledge Management*, which has been the use of 15 out of 35 ontologies (42.86%). In this category, ontologies have been used to support knowledge capture, representation, organization, use, sharing, among others. In this context, #06, #07, #12, #14, #23, #26, #27 and #33 were used to structure knowledge bases. For example, #06 was used in a recommender system supporting HCI design for web applications; #07 served as a semantic representation to an interactive system, enabling the UI to adapt itself to each user; #23 structured knowledge in an Evolutionary Algorithm approach, which combines Artificial Intelligence and design methods to automate development and improvement of web UI; #27 was used in a tool to support automatic evaluation of web UIs based on usability guidelines; and #33 was used as a basis of an affective multimodal conversational system. Some ontologies were used to knowledge capture and sharing. For example, #11 was used in an ontology-driven modeling framework to capture and manage design expert knowledge



Figure 5. Ontology-based solutions.

to be provided as design-related recommendations. #03 was used in a user's personal information management system to capture knowledge and adapt the UI according to the user impairments. By using a template to capture knowledge, #19 supported reverse engineering requirements for systems without documentation or available source code. #14 was used to increase the performance of user task detection by capturing simple interaction events (key pressed and mouse clicks) from the user's computer desktop which are automatically stored into a repository for further inferences. The other three ontologies (#04, #21, #32) were used for knowledge representation. #04 regards the representation of social context and its surrounding activities to support the development of mobile service provision models, #21 represents knowledge about the user's interaction events in different social media and #32 structures knowledge about web UI metrics.

Nine (25.71%) ontologies were used to *Reasoning*, i.e., to derive information that is not explicitly expressed in the ontology or in the knowledge base (Victoria et al., 2008). For example, #25 was used to determine the most suitable and adaptable user interfaces to craftswoman (user) considering modalities and sensory and motor abilities; #34 was used to build services that generate interaction tasks for social robots in smart IoT environments; and #12 was used to reason about the usability of interactive medical devices in multiple situations of use.

The third main use of HCI ontologies, with six out of 35 (17.14%) ontologies, has been *Basis of a System/Approach*, *Conceptual Description* and *Vocabulary/Communication*. Being a *Basis of a System/Approach* means that the ontology is used as part (e.g., a module or a repository) of the proposed system/approach architecture. This is the case, for example, of #29 and #34. The former is part of a context-aware computing architecture, supporting architecture layers such as knowledge discovery and context-aware services. The latter is part of the reasoning service component of a semantic-based platform architecture conceived as a middleware.

Ontologies used for *Conceptual Description* aim to provide a formal specification of a portion of the domain of interest. In this context, #18 provides a conceptualization regarding haptic interaction; #01 describes the HCI phenomenon, #2 conceptualizes interaction based on task description; #17 provides a conceptual representation about personas; #22 presents a conceptual description about user feedback; and #24 provides a semantic representation of concepts that are required for addressing universal accessibility related to user interaction with Information and Communication Technologies.

Ontologies used for *Vocabulary/Communication*, as the name suggests, are used as a vocabulary or for communication purposes. In this sense, #35 was used as a common vocabulary in an approach to

allow for automatic assessment of the graphical UI prototypes against user stories. #04 supported communication related to social context and its surrounding activities to aid the development of advanced models of mobile service. #21 represents user interaction events in different social media and it was used as a vocabulary for semantic event matching. #13 helped to inform the design of service-oriented architectures for engineering new systems and applications. #20 was used to support building user profiles. Finally, #28 served as a consensual instrument to talk about pervasive games from the user experience perspective.

Five out of 35 (14.29%) ontologies were used to *Semantic Mapping*, i.e., interrelate concepts from different sources (e.g., data bases, documents, ontologies, etc.). For instance, #31 was used to map concepts related to virtual reality (VR) system and VR environment, enabling the proposed system to assign interaction channels to the interaction capabilities of the VR environment. Finally, five (14.29%) ontologies were used to *Semantic Annotation*, meaning that semantic information was added to artifacts, allowing machines to read them. For example, in #27, each usability guideline was annotated enabling information to be accessed in a machine-processable form to automatically access the UI elements against selected UI guidelines and the associated metrics.

Table 3 summarizes the aspects addressed by each HCI ontology (RQ2) and the ontology-based solutions on which the ontologies were used (RQ3).

When extracting data to answer RQ3, we also extracted information about the benefits of using ontologies in the HCI domain. Table 4 summarizes the benefits reported in the publications and the ontologybased solutions they related to.

**Ontology Development (RQ4)**: According to the publications, the development of only three (8.57%) out of 35 ontologies followed an ontology engineering method (see Figure 6): #25 adopted Methontology (Fernández-López et al., 1997), #28 used Methontology (Fernández-López et al., 1997) and NeOn (Gómez-Pérez and Suárez-Figueroa, 2009; Suárez-Figueroa et al., 2012) and #30 applied NeOn (Gómez-Pérez and Suárez-Figueroa, 2009; Suárez-Figueroa et al., 2012). Although other publications (#02, #11, #12, #13 and #18) mention the use of some guidelines to develop the ontologies, the methods are not clearly presented or referred to. Thus, we did not consider that they adopted an ontology engineering method.



Figure 6. Ontology development.

**Ontology Evaluation (RQ5)**: In this research question, we used the classification proposed by Brank et al. (2005), namely: task-based evaluation and data-driven evaluation. In a task-based evaluation, the ontology is used in some kind of application or task, and the outputs of the application, or its performance on the given task, are used to evaluate the ontology (Brank et al., 2005). This is an indirect way of evaluating an ontology and, depending on the complexity or subjectivity of the application, it is hard to say if the ontology is really correct. In a data-driven evaluation, the ontology is compared to existing data about the problem domain to which the ontology refers (Brank et al., 2005). As shown in Figure

HCI aspects covered						Ontology-based solution adopted								
Ы	Ш	HCI	Pervasive	User Modeling /	HCI	Interaction	Adaptive Interactive	КM	Ressoning	Basis of a	Conceptual	Vocabulary	Semantic	Semantic
IU	01	phenomenon	Computing	Profile	Design	Experience	System	IXIVI	Reasoning	System	Description	vocabulary	Mapping	Annotation
#01		$\checkmark$									$\checkmark$			
#02					$\checkmark$						$\checkmark$		$\checkmark$	
#03	$\checkmark$			$\checkmark$		$\checkmark$	$\checkmark$	$\checkmark$						
#04	$\checkmark$		$\checkmark$	$\checkmark$				$\checkmark$				$\checkmark$		
#05			$\checkmark$						$\checkmark$					
#06					$\checkmark$			$\checkmark$						
#07				$\checkmark$			$\checkmark$	$\checkmark$						
#08	$\checkmark$				$\checkmark$			$\checkmark$					√	
#09	$\checkmark$													
#10	$\checkmark$				,			,		√				
#11			/		$\checkmark$			$\checkmark$	,	$\checkmark$				
#12		/	$\checkmark$		/			$\checkmark$	$\checkmark$			/		
#15		V	/					/				V		
#14	/	V	V					V						/
#15	V			/	/									V
#10				V	V						./			V
#18		./		v	V				./		v			
#19	1	V						./	v		v			
#20	v	v			./	./		v				./		
#21					v	v		1				Ň		
#22		•						v				•		
#23											•			
#24			$\checkmark$	$\checkmark$		$\checkmark$					$\checkmark$			
#25	$\checkmark$	$\checkmark$		$\checkmark$		$\checkmark$			$\checkmark$					
#26	$\checkmark$			$\checkmark$		$\checkmark$		$\checkmark$	$\checkmark$	$\checkmark$				
#27						$\checkmark$		$\checkmark$	$\checkmark$					$\checkmark$
#28			$\checkmark$			$\checkmark$						$\checkmark$		
#29			$\checkmark$						$\checkmark$	$\checkmark$				
#30				$\checkmark$		$\checkmark$			$\checkmark$					
#31	,	$\checkmark$	$\checkmark$		,			,		,				
#32	$\checkmark$		,		√			$\checkmark$		√				
#33		$\checkmark$	$\checkmark$	/				$\checkmark$	,					/
#34	/	V	$\checkmark$	$\checkmark$					$\checkmark$			1		$\checkmark$
#35	$\checkmark$	$\checkmark$										$\checkmark$		

Table 3: HCI ontologies: HCI aspects and ontology-based solutions.

	Benefits of using ontologies in the HCI area		Ontology-based solution adopted						
Id	Main Benefits	КМ	Reasoning	Basis of a System	Conceptual Description	Vocabulary	Semantic Mapping	Semantic Annotation	
#01	Describe the HCI phenomenon				Ŵ				
#02	Acquire a task-based description of interaction				v				
#03	Customize / adapt the UI according to user impairments	$\checkmark$							
#04	Support user stereotypes through the concept of social roles to research on content negotiation and deployment of different interaction styles	V				$\checkmark$			
#05	Make pervasive computing environments smarter with multimodal system and services		V						
#06	Aid recommender system to support interface design for modern applications and websites	1	v						
#07	Assist in adapting interactive systems to user behavior	V							
#08	Support multiple and different views/compositions of Web applications according to a dataset	v v					1	1	
#09	Engineer accessible web applications	v					N/	v	
#10	Assist user interface model for generating graphical user interface						V V		
#11	Support designers' activities	1		Ň			•		
#12	Aid safety-critical interactive devices and their use within multiple locations	v	V	v					
#13	Assist in informing the design of Service-oriented Architectures for controlling smart homes with gesture commands	v	•			V			
#14	Automatic detect the tasks a user is performing on his/her interaction	√				v			
#15	Support designers' activities	, v							
#16	Link Persona and Usability test methods in User-Centered Design							v v	
#17	Support personas to be effective tools for design							v	
#18	Design effective user interfaces and assist in the development of software modeling for haptic devices		√		, V				
#19	Recover requirements and reconstruct the software requirements document through observing the user interacting with	V	•		v				
	the system	, v							
#20	Recommend custom interfaces based on user profiles								
#21	Capture users' interaction activities in different social media	$\checkmark$				v			
#22	Improve User Interface adaptation through user's feedback								
#23	Support automated development and improvement of web user interfaces								
#24	Express personal user needs and preferences across ICT								
#25	Support suitable and adaptable interfaces for the user								
#26	Aid customization of user interface for people with disabilities	$\checkmark$	, V						
#27	Assist Web user interface (UI) automatic evaluation		V						
#28	Serve as a vocabulary of pervasive games from the user experience perspective		•			1		•	
#29	Support context-aware adaptation in mobile applications								
#30	Assist recommender system to suggest software adaptations related to user characterization		v	•					
#31	Build resilient virtual reality (VR) applications that can be automatically adapted to a particular VR system configura- tion						$\checkmark$		
#32	Support assessing Web User Interfaces using metrics from different providers								
#33	Guide the development of affective resources and/or applications	V		V					
#34	Build services that generate interaction tasks for social robots in smart IoT environments			•					
#35	Perform automatic assessment of the Graphical User Interfaces prototypes against User Stories					$\checkmark$		•	

# Table 4: Benefits of using ontologies in the HCI domain and ontology-based solution.

7, 15 (42.86%) ontologies (#02, #05, #06, #07, #08, #10, #12, #14, #15, #16, #21, #26, #29, #31, #35) were evaluated by following a task-based evaluation, two (5.71%) ontologies (#09, #30) used a datadriven evaluation method, five (14.29%) ontologies (#18, #19, #25, #27, #33) used both methods and 13 (37.14%) ontologies (#01, #03, #04, #11, #13, #17, #20, #22, #23, #24, #28, #32, #34) were not evaluated.



Figure 7. Ontology evaluation.

20 (51.14%) ontologies (#02, #05, #06, #07, #08, #10, #12, #14, #15, #16, #18, #19, #21, #25, #26, #27, #29, #31, #33, #35) were evaluated through a task-based method. In this kind of evaluation, ontologies are used to build solutions that are objects of evaluation. Thus, the ontology is indirectly evaluated. For example, the evaluation of #25 and #31 was carried out considering prototypes built using the ontologies. Analogously, systems built using the ontologies were used to evaluate #06 (university website and recommender system), #07 (interactive dashboard), #12 (safety-critical system), #26 (hospital device system) and #27 (interface customization system). #15 was evaluated through a RESTful Web Service that was used to deploy, generate and retrieve RDFa annotated web forms using the ontology. In #16, a semantic HTML template was applied in a small-scale experiment to build a web-based application for managing academic information (e.g., grades, schedule). #19 was evaluated by means of a template used to recover system requirements from the actions performed by the user.

Seven (17.5%) ontologies (#09, #18, #19, #25, #27, #30, #33) adopted a data-driven approach. All of them used some kind of instantiation to evaluate the ontology. For example, #19 was instantiated considering a real case of requirements recovery, while #27 was evaluated through the instantiation of 115 usability guidelines. #30 was the only ontology evaluated by using a data-driven approach and that used competency questions to represent the ontology requirements. In its evaluation, #30 instantiation was oriented to the competency questions.

Finally, 13 (37,14%) ontologies (#01, #03, #04, #11, #13, #17, #20, #22, #23, #24, #28, #32, #34) were not evaluated. In some cases, the authors stated that the ontology was under evaluation (e.g., #24). In others, toy examples or workshops were used to illustrate the ontology use (e.g., #11, #34). Based on the information provided in the publications, we did not consider that these ontologies were evaluated because they were not actually used in real situations nor instantiated real data.

**Ontology Quality Characteristics (RQ6)**: in this research question, we considered some of the characteristics of "beautiful ontologies" defined by D'Aquin and Gangemi (2011)<sup>9</sup>, namely: (i) reusing foundational ontologies; (ii) being formally rigorous; (iii) implementing also non-taxonomic relations; (iv)

<sup>&</sup>lt;sup>9</sup>We did not consider characteristics such as commercial impact, which are difficult to evaluate based only on information provided in the publications.

being modular or embedded in a modular framework; (v) implementing an international standard; (vi) being based on competency questions, and (vii) following an evaluation method. Table 5 presents the quality characteristics identified in the ontologies.

Research questions and their rationale.										
id	reusing foundational ontologies	being formally rigorous	implementing also non-taxonomic relations	being modular	implementing an international standard	based on competency questions	following an evaluation method			
#01			$\checkmark$							
#02		$\checkmark$	$\checkmark$				$\checkmark$			
#03			$\checkmark$							
#04			$\checkmark$	$\checkmark$	$\checkmark$					
#05		$\checkmark$	$\checkmark$	$\checkmark$			$\checkmark$			
#06		$\checkmark$	$\checkmark$				$\checkmark$			
#07		$\checkmark$	$\checkmark$	$\checkmark$			$\checkmark$			
#08			$\checkmark$	$\checkmark$			$\checkmark$			
#09			$\checkmark$				$\checkmark$			
#10			$\checkmark$				$\checkmark$			
#11		$\checkmark$	$\checkmark$	$\checkmark$		$\checkmark$				
#12		$\checkmark$	$\checkmark$			$\checkmark$	$\checkmark$			
#13			$\checkmark$			$\checkmark$				
#14				$\checkmark$			$\checkmark$			
#15										
#16							$\checkmark$			
#17										
#18										
#19			$\checkmark$				$\checkmark$			
#20										
#21		$\checkmark$	$\checkmark$				$\checkmark$			
#22										
#23		,	,		,					
#24		$\checkmark$	$\checkmark$	$\checkmark$			,			
#25				1						
#26		$\checkmark$	$\checkmark$	$\checkmark$			$\checkmark$			
#27						1	$\checkmark$			
#28			$\checkmark$			$\checkmark$	1			
#29		1	1	1		/	$\checkmark$			
#30		$\checkmark$	$\checkmark$	$\checkmark$		$\checkmark$	$\checkmark$			
#31			1		1		$\checkmark$			
#32		1	$\checkmark$	$\checkmark$	$\checkmark$		/			
#35 #24	/		$\checkmark$		/		$\checkmark$			
#34	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$		/			
#33							$\checkmark$			

Table 5 esearch questions and their ration

Concerning the reuse of foundational ontologies, only #18 and #34 (5.71%) were developed based on

a foundational ontology (Basic Formal Ontology –  $BFO^{10}$  (Grenon et al., 2004) and Dolce Ultralight upper ontology<sup>11</sup> (Gangemi et al., 2002), respectively).

With respect to being formally rigorous, we considered two aspects: whether axioms/rules are defined, and how the ontology is presented (as a graphical conceptual model, as an operational ontology, using some formalism, or textual description only). Eight (22.86%) ontologies define rules (#02, #05, #06, #07, #11, #24, #25, #26), two (5.71%) ontologies define axioms (#30, #33) and four (11.43%) ontologies define both (#12, #18, #21, #34). Thus, 14 ontologies (40%) present some degree of formalism. Only one (7.14%) of those 14 ontologies (#02) is presented by means of textual description (since it defined rules, we consider that it presents some degree of formalism). Two (14.29%) (#07, #18) are operational ontologies, being represented as OWL and OWL SWRL artifacts, respectively. Five (35.71%) ontologies (#06, #11, #21, #24, #26) are presented as graphical conceptual models. The remaining six (42.86%) ontologies (#05, #12, #25, #30, #33, #34) are presented as both, graphical conceptual model and operational ontology.

Regarding the types of relations, 27 (77.14%) ontologies consider both taxonomic and non-taxonomic relations (although #01, #10 do not clearly represent them). Six (17.14%) ontologies (#14, #15, #20, #23, #29, #31) consider only taxonomic relations. In two (5.71%) ontologies (#22, #35) it was not possible to identify the types of relations.

As for modularity, we investigated if the ontology was organized in hierarchy, dimension, abstraction levels, structure of classes, or if it is non-modular. 12 (34.29%) ontologies are non-modular, while 23 (65.71%) ontologies are modularized according to one of the aforementioned modularization types.

Concerning the use of standards, only four (11.43%) ontologies mention their use as a reference to the ontology conceptualization: #04 used the Composite Capabilities/Preference Profiles part of the Device Independence activity of W3C (Klyne et al., 2004), #24 considered ISO/IEC 24751:2008 (ISO, 2008), ETSI TS 202 746 (ES, 2010) and ISO 9999:2011 (ISO, 2011), #32 used W3C Web Content Accessibility Guidelines<sup>12</sup> and #34 considered the vcard specification (RFC6350<sup>13</sup>) and the W3C Organization Ontology<sup>14</sup>. Other works (#01, #02, #03, #10, #13, #16, #17, #18, #28, #33) mention the use of references but do not specify which ones were used. #28 and #33 mention models found in the literature. Therefore, we did not consider that they use international standards.

Regarding the use of competency questions, only five (14.29%) ontologies (#11, #12, #13, #28 and #30) used competency questions to establish the ontology requirements.

# 5. Data Analysis and Discussion

In this section, we provide additional information about the analyzed HCI ontologies and make some discussion about the obtained results.

Considering *when the studies have been published*, HCI ontologies only became a regular research topic addressed in publications from 2010. We believe that this is due to the increasing use of ontologies in several domains in the last decade (Jean-Baptiste, 2021) (ontologies became particularly popular

<sup>&</sup>lt;sup>10</sup>http://ontology.buffalo.edu/bfo/

<sup>&</sup>lt;sup>11</sup>http://www.ontologydesignpatterns.org/ont/dul/DUL.owl

<sup>12</sup>https://www.w3.org/WAI/ER/tools/

<sup>13</sup>https://www.w3.org/TR/vcard-rdf/

<sup>14</sup>https://www.w3.org/TR/vocab-org/

because of Semantic Web) and the acknowledgment of the advantages of using them to solve knowledgerelated problems. Regarding the peak of publications, in our study it appears in 2012 (7 studies), while in the study by Tarhan and Giray (2017) (2005-2016) it appears in 2008 (3 studies) and 2012 (3 studies), in the work by Tapia-Leon et al. (2018) (2012-2017) it appears in 2013 (14 studies) and 2014 (13 studies) and in the work by Sousa et al. (2016) (2006-2014) it appears in 2014 (4 studies) followed equally by 3 studies in 2006, 2011 and 2013.

Regarding the *type of vehicle*, we noticed a predominance of conference papers (26 out 38 papers, 68.42%). The shares in our study are close to the ones found in secondary studies investigating the use of ontologies in other domains (Sousa et al., 2016; Tapia-Leon et al., 2018; Tarhan and Giray, 2017), in which conference represents an average of 64% of the papers. Considering studies that investigate other aspects in the Computer Science field, the shares in our study are close to the ones found by Salamon et al. (2019), Quirino et al. (2018) and Real et al. (2019). Taking the period of publications into account, we noticed that the investigated topic has been target of researchers for more than 25 years. However, there is a predominance of conference papers, which, as in the aforementioned studies, can be seen as an indication that the research on this topic is not very mature yet, since journal publications usually require more mature works. We also noticed that there is no "home" or well-established forum for publications addressing HCI ontologies, since the 38 papers were published in many different conferences and journals. In fact, it seems that papers addressing HCI ontologies have been more welcome in vehicles of areas such as Information Technology and Artificial Intelligence than in vehicles devoted to HCI.

With respect to the HCI aspects covered by the ontologies, the top two aspects have been User Interface (12 ontologies) and HCI Phenomenon (11 ontologies). These aspects concern core knowledge of the HCI domain. When talking about HCI, it is often necessary to talk about the HCI phenomenon itself and the parts involved in it, i.e., the user and the system, which is perceived by means of the user interface. Thus, it was expected that these aspects were the focus of many ontologies. *Pervasive Computing*, User Modeling / Profile and HCI Design have also been addressed by many ontologies (ten) followed by Interaction Experience (eight) and Adaptive Interactive System (three). This shows that representing knowledge related to the design of interactive computer system based on user needs offers the potential to achieve personalized interaction and adaptive user interface, promoting a better user experience in times of pervasive systems. Although they are different aspects, there is an interrelation when we look at them in a holistic way. The evolution of one contributes to the evolution of the others, in a virtuous cycle, reflecting the core concerns of the HCI area and motivating the emergence of new technologies and studies for the benefit of people's daily lives. Even though there are ontologies that have addressed these aspects, there are some topics that have not been much explored (e.g., user feedback, usability) and others that have not even been considered (e.g., evaluation methods, prototype), suggesting that there are still many portions of the HCI domain that can be addressed by ontologies.

Considering when the studies have been published and HCI aspects covered by the ontologies we noticed that, although several HCI aspects have been addressed (Figure 4), the target of interest has changed over the years. For example, User Interface is the most addressed aspect in the first period (1994-2010), when ontologies started to be applied in HCI area. In the 2011-2015 period, which has the highest number of publications, HCI Design stands out, while in the 2016-2019 period, Pervasive Computing, User Modeling / Profile and Interaction Experience are the aspects addressed by more publications. We believe that, at first, ontologies were used to address more basic and perceptible aspects (e.g., User Interface) and, as the use of ontologies as a whole got more mature, other aspects were also addressed. The evolution and change of concerns in the HCI area can also have contributed to that. Despite the little use of ontologies in the first half of the 26 years considered in the study, the use of ontologies in

the last decade reflects (in some way or to some degree) recent concerns in the area of HCI. Moreover, the increasing number of ontologies, the fact that HCI ontologies have addressed a wider range of HCI aspects associated with the increasing number of ontologies represented as both conceptual and computational models (which increases the range of possible ontology-based solutions) can be understood as an indication that the use of ontologies in HCI has been target of interest by the HCI community.

Considering how ontologies have been used in the HCI domain, there has been a predominance of knowledge management solutions (15 ontologies) and reasoning (9 ontologies). Ontologies are useful to structure data and knowledge (Jean-Baptiste, 2021). In this study, knowledge management encompasses issues such as knowledge representation, structuring, organization and sharing. As ontologies primarily serve to the representation purpose, it is not a surprise that most publications appear in this category. Knowledge management has been an ontology use of paramount importance in several domains (de Souza et al., 2015). The study results showed that the HCI domain has also benefited from the use of ontologies in this kind of application. Ontologies as conceptual models as well as operational ontologies have been used in knowledge management solutions. On the other hand, reasoning has involved the use of operational ontologies to support getting new information from data captured by an application or stored in a repository. By combining results from RQ2 and RQ3, we can notice that the two more used ontology-based solutions (knowledge management and reasoning) have been used only in the context of the four more addressed HCI aspects (UI, HCI phenomenon, User Modeling / Profile and HCI design). In line with what we said before, we believe that these aspects have been much explored because they are core aspects in the HCI domain. Thus, we think that it would be expected that they were addressed in the ontology-based solutions most used in the HCI domain. Although interactive computer system and user interface evaluation is an important aspect of the HCI domain (Preece et al., 2015), we noticed that it has not been much explored.

By analyzing the ontologies use and benefits reported in the selected publications, we can point out some advantages that have been obtained from the use of ontology-based solutions to solve HCI problems: (i) by using ontologies it has been possible to represent knowledge about the domain in a clear and explicit way, supporting communication and learning tasks. For example, #01 and #28 describe and make clear knowledge about HCI phenomenon and pervasive games, respectively. (ii) ontologies have allowed structuring knowledge, supporting the development of ontology-based systems, frameworks, approaches, architectures, etc. For example, #07 is used (both at design time and at runtime) in a framework for adapting interactive systems according to the user behavior. (iii) ontologies have enabled automated solutions and reasoning. For instance, #12 is used to reason about the usability of interactive medical devices in multiple situations of use. (iv) ontologies have supported semantic interoperability, enabling data/information from different sources to be integrated into HCI solutions (e.g., #16 it is used to semantically annotate a template about persona to integrate multiple data).

Concerning the *quality of HCI ontologies*, we noticed that although the ontologies present some quality characteristics (e.g., being modular and implementing non-taxonomic relations), others have not been a concern. For example, only one ontology (#34) is grounded in a foundational ontology. Foundational ontologies provide basic concepts and help domain ontologies to be truthful. Most of the HCI ontologies are not formally rigorous. For example, the ontologies presented in #13, #15, #20, #23, #29 and #31 are, in fact, taxonomies or lightweight ontologies, while #07, #08, #10 and #18 are just OWL artifacts (i.e., operational ontologies). Being formally rigorous makes the ontology more precise. As it was expected, Web Ontology Language (OWL) and Resource Description Framework (RDF) were the main languages used. This demonstrates that the publications may be concerned with using a language recommended by the World Wide Web Consortium (W3C) to specify and manipulate ontologies. Only four ontologies

used international standards as a basis. Standards are important references to ontologies because they provide a consensual conceptualization of the domain of discourse. As for ontology development, only three were developed by following an ontology engineering method. By following an ontology engineering method, one adopts best practices to increase ontology quality. In this context, only three ontologies had their scope defined by means of competency questions. These questions help to elicit the ontology requirements and produce an ontology that meets the user's needs. Most HCI ontologies have been evaluated. However, evaluation has not been an explicit subject in the publications. Most of the times we considered that the instantiations or uses made of the ontologies were a way of evaluating them. The quality of an ontology directly affects the solution built using it. Thus, it is of paramount importance to use good practices to develop HCI ontologies. In general, the study results showed that ontology engineering practices and principles have not been a concern when developing HCI ontologies. We think that this can be seen as an opportunity for HCI and Ontology Engineer professionals and researchers to get closer.

Finally, by analyzing the ontologies coverage (we extracted the ontologies concepts, as it can be seen in our SLR package (Costa et al., 2020)), we noticed several overlaps. Although several ontologies address the same concepts or complementary portions of the HCI domain, they do not reuse each other's concepts. In fact, we noticed that HCI ontologies have been developed for solving specific problems, in specific contexts, without concern with integration to other existing HCI ontologies. This causes an increasing number of ontologies and makes it difficult to integrate different ontologies to produce a new ontology covering a larger portion of the domain. This, in turn, leads one to create new ontologies instead of reusing existing ones, feeding a 'vicious cycle' of many overlapping ontologies.

# 6. Limitations of the Study

Although the ambition of a systematic literature review is to summarize all relevant research in an area, different sets of publications can be obtained given a number of decisions and judgments taken (Wohlin et al., 2012). Thus, as any study, the study presented in this paper has some limitations, which must be considered together with the results. In addition, some challenges can reach the researchers during a systematic literature review, such as how to select a comprehensive and relevant source of publications, how to consistently apply the inclusion/exclusion criteria, how to classify data and how to interpret them. In this study, we experienced these challenges and carried out some actions aiming at minimalizing their influence on the results.

Publication selection and data extraction were initially performed by one of the authors. To reduce this subjectivity, the other two authors performed these same steps over a random sample of about 45% of the non-duplicated publications. Thus, at least 400 publications were evaluated by two researchers. The results of each reviewer were then compared. Discordances and possible biases were discussed in meetings. Concerning the 38 selected publications, all of them were evaluated by at least two researchers. An analysis of the degree of concordance was performed to measure the level of agreement between the results obtained from the researchers in the selection process. For this, we calculated the kappa coefficient (Landis and Koch, 1977). Kappa is a statistical measure of inter-rater agreement for qualitative (categorical) items. It is considered a more robust measure than simple percent agreement calculation, since it takes into account the agreement occurring by chance. Kappa maximum value is 1, representing total agreement. Values close to and below 0 indicate no agreement. Considering the set of publications analyzed by all the reviewers, the obtained kappa coefficient was 0.79, which, according to Landis and Koch (1977), means substantial agreement.

Terminological problems in the search strings may have led to missing some primary studies. In order to minimize these problems, we performed previous simulations in the selected databases. We decided not to search any specific conference proceedings, journals, or the grey literature (technical reports and works in progress). Thus, we have just worked with studies indexed by the selected electronic databases. The exclusion of these other sources makes the review more repeatable, but possibly some valuable studies may have been left out of our analysis. Although the digital libraries adopted represents a comprehensive source of publications, the exclusion of other sources and the fact that we did not perform snowballing may have left some valuable publications out of our analysis.

Applying the search string to the digital libraries engine can also interfere with the returned publications. First, because it is necessary to adapt the string syntax. The adaptation may result in a different search string. The string used in the study was adapted to each digital library and we double-checked the adapted string and ran tests to ensure that it was correct. Second, because of problems or changes in the engines. During the study, changes occurred in the engine of one of the used digital libraries. We ran some tests that showed us that the engine interface had changed, but the engine seemed to work as before.

Another limitation is related to the classifications we made for categorizing data. Some categories were based on classifications proposed in the literature (e.g., quality characteristics (D'Aquin and Gangemi, 2011), evaluation types (Brank et al., 2005)). Others were established during data extraction, based on data provided by the analyzed publications (e.g., categories used in RQ2 and RQ3). Classification schemas and data categorization were done by the first author and reviewed by the other two authors. However, determining how publications fit them involves a lot of judgment. Thus, different results could be obtained by other researchers.

# 7. Conclusions

In this paper, we presented a systematic literature review that investigated ontologies in the HCI domain. The results of this study provide a panorama of research related to the topic and go deeper in some aspects about the use of ontologies to solve HCI-related problems. A total of 899 publications were considered and 35 ontologies addressing different aspects of the HCI domain were identified in 38 selected publications. Six research questions were defined to investigate the following facets: (i) the distribution of the selected publications over the years and the type of vehicle; (ii) the HCI domain aspects addressed by the HCI ontologies; (iii) how ontologies have been used in the HCI domain; (iv) how HCI ontologies have been developed; (v) how HCI ontologies have been evaluated; and (vi) the HCI ontology quality characteristics.

According to Kitchenham and Charters (2007), the results of a systematic literature review help identify gaps that suggest future research. Therefore, this study contributes by providing a consolidated view of ontologies in HCI and pointing out some conclusions that can drive future research in the area. In this context, we can highlight: (i) The adoption of ontologies in the HCI domain is a research topic that still needs to be matured. Although the first publication found in this study is from 1994, only in the last decade ontology use became more regular in the HCI domain. (ii) Ontologies have focused on core HCI aspects, mainly on the interface and interaction. This suggests that several HCI sub-domains have not experienced the benefits of ontology-based solutions. For instance, in most cases, there are topics of the HCI aspects that have not been much explored (e.g., user experience, usability) and others that have not even been considered (e.g., prototype, brain computer interface). Moreover, although some works deal

446

with HCI design and evaluation, there is still much to be explored. For example, we did not find any work addressing design or evaluation processes, documentation and evaluation methods. We also did not find any publication dealing with physiological computing, multiple users or smart home. (iii) Ontologies have been used mainly in knowledge management and reasoning solutions. Considering the diversity of knowledge involved in the HCI domain and the multiple artifacts used in this context, other important solutions, such as the ones related to semantic interoperability, can be further explored. (iv) The most sophisticated ontology-based solutions we found focus only on interface adaptation. There are many other HCI-related problems that can be addressed by using ontologies. (iv) There has been no concern with the quality of the developed ontologies. This suggests that HCI and ontology engineering professionals and researchers should work together to produce HCI ontologies. This can contribute to both areas: HCI can benefit from quality ontologies, while Ontology Engineering can find new opportunities and challenges to apply ontologies. (v) There has been a lack of reusing existing ontologies. Therefore, ontologies reuse in the HCI domain is an issue to be further explored. In summary, the study results show that although ontologies have been used in the HCI domain, the use has focused on only two types of ontology-based solutions and three core aspects of HCI. Moreover, HCI ontologies, in general, have not been developed by following ontology engineering good practices. Considering the advantages of using ontologies (as discussed in Section 6), we believe that it is worth investing efforts to improve HCI ontologies and ontology-based solutions to address problems in the HCI domain. Towards this direction, we propose to organize HCI ontologies as an ontology network to provide comprehensive and consistent knowledge of the HCI domain and support ontology-based solutions.

Concerning related works, we did not find any study investigating the use of ontologies in the HCI domain. Before performing our study, we searched for secondary studies investigating ontologies in the HCI domain. Since we did not find any, we decided to carry out our study. Therefore, there is a possibility that this study is a pioneering work investigating HCI ontologies.

As for the publications analyzed in this study, many times we faced difficulties to extract information about the ontologies, because it was incomplete or implicit in the paper. This makes it difficult to properly understand existing ontologies and hampers reuse them. We believe that the HCI community could achieve more benefits if the proposed ontologies were presented completely (even if through supplementary material). Another fundamental issue is to improve the visibility and ability to download ontologies and their documentation through the web.

Reuse is an important issue in ontology engineering and a desirable quality attribute of ontologies. Considering seven ontologies (#25, #27, #30, #32, #33, #34, #35) published in the last years and represented as both conceptual and computational models (which may contribute to reuse), we noticed that four of them were reused: #27, #32 and #34 in other works by the same authors; #30 and #34 by other authors. Since this is a small sample of the ontologies identified in our study, we cannot make conclusions about reuse considering only these ontologies. Thus, as future work, we intend to extend our study to investigate ontology reuse in HCI.

Finally, it is worth saying that, due to the successful use of ontologies in other areas, in this study, our focus was to investigate ontologies in HCI and verify if their potential has been explored to solve semantic and knowledge-related problems in that domain. Sometimes, ontologies can be used as taxonomies, but the opposite does not apply. Thus, since taxonomies can also be used to represent knowledge in the HCI area (e.g., (Dankov and Bontchev, 2020; Janssen et al., 2020; Jonas et al., 2019)) and support semantic and knowledge-based solutions, we consider in the future to investigate the use of taxonomies in HCI and compare the results with our study.

# References

- Arango-López, J., Vela, F.L.G., Collazos, C.A. & Moreira, F. (2018). Modeling and Defining the Pervasive Games and Its Components from a Perspective of the Player Experience. In Á. Rocha, H. Adeli, L.P. Reis and S. Costanzo (Eds.), Proceedings of the Trends and Advances in Information Systems and Technologies. WorldCIST'18 2018. Advances in Intelligent Systems and Computing, vol 746 (pp. 625–635). Cham: Springer International Publishing. doi:10.1007/978-3-319-77712-2\_58.
- Bakaev, M. & Avdeenko, T. (2010). Ontology to Support Web Design Activities in E-Commerce Software Development Process. In O.I. Potaturkin and Y.I. Shokin (Eds.), Proceedings of the IASTED International Conference on Automation, Control, and Information Technology Information and Communication Technology, ACIT-ICT 2010 (pp. 241–248). Novosibirsk, Russia: ACTA Press. doi:10.2316/p.2010.691-075.
- Bakaev, M. & Avdeenko, T. (2012). User Interface Design Guidelines Arrangement in a Recommender System with Frame Ontology. In H. Yu, G. Yu, W. Hsu, Y.-S. Moon, R. Unland and J. Yoo (Eds.), *Proceedings of the Database Systems for Advanced Applications. DASFAA 2012. Lecture Notes in Computer Science, vol 7240* (pp. 311–322). Berlin, Heidelberg: Springer Berlin Heidelberg. doi:10.1007/978-3-642-29023-7\_31.
- Bakaev, M. & Gaedke, M. (2016). Application of evolutionary algorithms in interaction design: From requirements and ontology to optimized web interface. In *Proceedings of the 2016 IEEE NW Russia Young Researchers in Electrical and Electronic Engineering Conference (EIConRusNW)* (pp. 129–134). St. Petersburg, Russia: IEEE. doi:10.1109/EIConRusNW.2016.7448138.
- Bakaev, M., Heil, S., Perminov, N. & Gaedke, M. (2019). Integration Platform for Metric-Based Analysis of Web User Interfaces. In M. Bakaev, F. Frasincar and I.-Y. Ko (Eds.), *Proceedings of the International Conference on Web Engineering. ICWE 2019. Lecture Notes in Computer Science, vol 11496* (pp. 525–529). Cham: Springer International Publishing. doi:10.1007/978-3-030-19274-7\_39.
- Bezold, M. & Minker, W. (2010). A Framework for Adapting Interactive Systems to User Behavior. Journal of Ambient Intelligence and Smart Environments, 2(4), 369–387. doi:10.3233/AIS-2010-0081.
- Bowen, J. & Hinze, A. (2012). Using Ontologies to Reason About the Usability of Interactive Medical Devices in Multiple Situations of Use. In *Proceedings of the 4th ACM SIGCHI Symposium on Engineering Interactive Computing Systems*. EICS '12 (pp. 247–256). Copenhagen, Denmark: ACM. doi:10.1145/2305484.2305525.
- Brank, J., Grobelnik, M. & Mladenić, D. (2005). A survey of ontology evaluation techniques. In Proceedings of the Conference on Data Mining and Data Warehouses (SiKDD 2005) (pp. 166–170). Ljubljana, Slovenia. doi:10.1.1.101.4788.
- Carroll, J.M. (2014). Human Computer Interaction (HCI). In M. Soegaard and R.F. Dam (Eds.), *The Encyclopedia of Human-Computer Interaction* (2nd ed., pp. 21–61). Aarhus, Denmark: The Interaction Design Foundation. Chapter 2.
- Celino, I. & Corcoglioniti, F. (2010). Towards the Formalization of Interaction Semantics. In *Proceedings of the* 6th International Conference on Semantic Systems. I-SEMANTICS '10 (pp. 10:1—10:8). Graz, Austria: ACM. doi:10.1145/1839707.1839719.
- Chera, C.-M., Tsai, W.-T. & Vatavu, R.-D. (2012). Gesture ontology for informing Service-oriented Architecture. In Proceedings of the 2012 IEEE International Symposium on Intelligent Control (pp. 1184–1189). Dubrovnik, Croatia: IEEE. doi:10.1109/ISIC.2012.6398257.
- Coronato, A. & De Pietro, G. (2007). A context service for multimodal pervasive environments. In Proceedings of the 2007 Third International Conference on Security and Privacy in Communications Networks and the Workshops - SecureComm 2007 (pp. 84–87). Nice, France, France: IEEE. doi:10.1109/SECCOM.2007.4550312.
- Costa, C. & Murta, L. (2013). Version control in Distributed Software Development: A systematic mapping study. In *Proceedings of the IEEE 8th International Conference on Global Software Engineering, ICGSE 2013* (pp. 90–99). doi:10.1109/ICGSE.2013.19.
- Costa, S.D., Barcellos, M.P., Falbo, R.d.A. & Castro (2020). Ontologies in HCI Protocol & Publications & Data Extraction. Available at https://bit.ly/\_hciR.
- Dankov, Y. & Bontchev, B. (2020). Towards a Taxonomy of Instruments for Facilitated Design and Evaluation of Video Games for Education. In *Proceedings of the 21st International Conference on Computer Systems and Technologies* '20 (pp. 285– 292). New York, NY, USA: ACM. doi:10.1145/3407982.3408010.
- D'Aquin, M. & Gangemi, A. (2011). Is there beauty in ontologies? *Applied Ontology*, 6(3), 165–175. doi:10.3233/AO-2011-0093.
- de Souza, É.F., Falbo, R.d.A. & Vijaykumar, N.L. (2015). Knowledge management initiatives in software testing: A mapping study. *Information and Software Technology*, 57, 378–391. doi:10.1016/j.infsof.2014.05.016.
- Devaurs, D., Rath, A. & Lindstaedt, S. (2012). Exploiting the user interaction context for automatic task detection. *Applied Artificial Intelligence*, 26(1-2), 58–80. doi:10.1080/08839514.2012.629522.
- Elyusufi, Y., Seghiouer, H. & Alimam, M.A. (2014). Building profiles based on ontology for recommendation custom interfaces. In Proceedings of the 2014 International Conference on Multimedia Computing and Systems (ICMCS) (pp. 558–562). Marrakech, Morocco: IEEE. doi:10.1109/ICMCS.2014.6911166.
- ES, E. (2010). Human Factors (HF); Personalization and User Profile Management; User Profile Preferences and Information.

- Ethier, J.-F., Barton, A. & Taseen, R. (2018). An ontological analysis of drug prescriptions. *Applied Ontology*, *13*, 273–294. doi:10.3233/AO-180202.
- Fernández-López, M., Gómez-Pérez, A. & Juristo, N. (1997). METHONTOLOGY: From Ontological Art Towards Ontological Engineering. In Proceedings of the Ontological Engineering AAAI-97 Spring Symposium Series. American Asociation for Artificial Intelligence. http://oa.upm.es/5484/.
- Fischer, G. (2001). User Modeling in Human–Computer Interaction. User Modeling and User-Adapted Interaction, 11(1), 65–86. doi:10.1023/A:1011145532042.
- Fonseca, V.S., Barcellos, M.P. & Falbo, R.d.A. (2017). An ontology-based approach for integrating tools supporting the software measurement process. *Science of Computer Programming*, 135, 20–44. doi:10.1016/j.scico.2016.10.004.
- Gangemi, A., Guarino, N., Masolo, C., Oltramari, A. & Schneider, L. (2002). Sweetening Ontologies with DOLCE. In A. Gómez-Pérez and V.R. Benjamins (Eds.), *Proceedings of the Knowledge Engineering and Knowledge Management: Ontologies and the Semantic Web* (pp. 166–181). Berlin, Heidelberg: Springer Berlin Heidelberg. doi:10.1007/3-540-45810-7\_18.
- Garay-Vitoria, N., Cearreta, I. & Larraza-Mendiluze, E. (2019). Application of an Ontology-Based Platform for Developing Affective Interaction Systems. *IEEE Access*, 7, 40503–40515. doi:10.1109/ACCESS.2019.2903436.
- Gómez-Pérez, A. & Suárez-Figueroa, M.C. (2009). NeOn Methodology for Building Ontology Networks: a Scenario-based Methodology. In Proceedings of the International Conference on Software, Services & Semantic technologies (S3T 2009). Bulgaria: Universidad de Sofia. http://oa.upm.es/5475/.
- Grenon, P., Smith, B. & Goldberg, L. (2004). Biodynamic Ontology: Applying BFO in the Biomedical Domain. Studies in health technology and informatics, 102, 20–38. doi:10.3233/978-1-60750-945-5-20.
- Gruber, T.R. (1995). Toward principles for the design of ontologies used for knowledge sharing? International Journal of Human-Computer Studies, 43(5-6), 907–928. doi:10.1006/ijhc.1995.1081.
- Gubert, L.C., da Costa, C.A. & Righi, R.d.R. (2020). Context awareness in healthcare: a systematic literature review. Universal Access in the Information Society, 19(2), 245–259. doi:10.1007/s10209-019-00664-z.
- Guerino, G.C. & Valentim, N.M.C. (2020). Usability and user experience evaluation of natural user interfaces: a systematic mapping study. *IET Software*, 14(5), 451–467. doi:10.1049/iet-sen.2020.0051.
- Guizzardi, G. (2007). On Ontology, Ontologies, Conceptualizations, Modeling Languages, and (Meta)Models. In O. Vasilecas, J. Edler and A. Caplinskas (Eds.), *Proceedings of the Frontiers in Artificial Intelligence and Applications, Databases* and Information Systems IV. Amsterdam, The Netherlands, The Netherlands: IOS Press. http://dl.acm.org/citation.cfm?id= 1565421.1565425.
- Haller, A., Groza, T. & Rosenberg, F. (2012). Interacting with Linked Data via Semantically Annotated Widgets. In J.Z. Pan, H. Chen, H.-G. Kim, J. Li, Z. Wu, I. Horrocks, R. Mizoguchi and Z. Wu (Eds.), *Proceedings of the The Semantic Web*, . *JIST 2011. Lecture Notes in Computer Science, vol 7185* (pp. 300–317). Berlin, Heidelberg: Springer Berlin Heidelberg. doi:10.1007/978-3-642-29923-0\_20.
- ISO (2008). Information technology Individualized adaptability and accessibility in e-learning, education and training Part 1: Framework and reference model. Standard ISO/IEC 24751-1:2008, International Organization for Standardization. https://www.iso.org/standard/41521.html.
- ISO (2011). Assistive products for persons with disability Classification and terminology. Standard ISO 9999:2011, International Organization for Standardization. https://www.iso.org/standard/50982.html.
- Janssen, A., Passlick, J., Rodríguez Cardona, D. & Breitner, M.H. (2020). Virtual Assistance in Any Context: A Taxonomy of Design Elements for Domain-Specific Chatbots. *Business & Information Systems Engineering*, 62(3), 211–225. doi:10.1007/s12599-020-00644-1.
- Jean-Baptiste, L. (2021). Ontologies with Python (p.344). Berkeley, CA: Apress. doi:10.1007/978-1-4842-6552-9.
- Jonas, M., Said, S., Yu, D., Aiello, C., Furlo, N. & Zytko, D. (2019). Towards a Taxonomy of Social VR Application Design. In Extended Abstracts of the Annual Symposium on Computer-Human Interaction in Play Companion Extended Abstracts (pp. 437–444). New York, NY, USA: ACM. doi:10.1145/3341215.3356271.
- Kabir, M.A., Han, J., Yu, J. & Colman, A. (2014). Inferring User Situations from Interaction Events in Social Media. *The Computer Journal*, 58(9), 2026–2043. doi:10.1093/comjnl/bxu131.
- Karim, S. & Tjoa, A.M. (2006). Towards the Use of Ontologies for Improving User Interaction for People with Special Needs. In K. Miesenberger, J. Klaus, W.L. Zagler and A.I. Karshmer (Eds.), *Proceedings of the International Conference on Computers for Handicapped Persons (ICCHP) Computers Helping People with Special Needs* (pp. 77–84). Berlin, Heidelberg: Springer Berlin Heidelberg. doi:10.1007/11788713\_12.
- Kitchenham, B.A. & Charters, S. (2007). Guidelines for performing Systematic Literature Reviews in Software Engineering. Technical report, Software Engineering Group School of Computer Science and Mathematics Keele University and Department of Computer Science University of Durham.
- Klyne, G., Reynolds, F., Woodrow, C., Ohto, H., Hjelm, J., Mark H., B. & Tran, L. (2004). Composite capability/preference profiles (cc/pp): structure and vocabularies 1.0. Technical report, W3C.

- Korfiatis, N. & Constantiou, I. (2006). Socially targeted mobile services: Towards an upper level ontology of social roles for mobile environments. In *14th European Conference on Information Systems* (p.12). Göteborg; Sweden.
- Koutkias, V., Kaklanis, N., Votis, K., Tzovaras, D. & Maglaveras, N. (2016). An Integrated Semantic Framework Supporting Universal Accessibility to ICT. Universal Access in the Information Society, 15(1), 49–62. doi:10.1007/s10209-014-0372-1.
- Kultsova, M., Potseluico, A., Zhukova, I., Skorikov, A. & Romanenko, R. (2017). A Two-Phase Method of User Interface Adaptation for People with Special Needs. In A. Kravets, M. Shcherbakov, M. Kultsova and P. Groumpos (Eds.), Proceedings of the Conference on Creativity in Intelligent Technologies and Data Science. CIT&DS 2017. Communications in Computer and Information Science, vol 754. (pp. 805–821). Volgograd; Russia: Springer, Cham. doi:10.1007/978-3-319-65551-2\_58.
- Landis, J.R. & Koch, G.G. (1977). The Measurement of Observer Agreement for Categorical Data. *Biometrics*, 33(1), 159–174. doi:10.2307/2529310.
- Lebib, F.-Z., Mellah, H. & Mohand-Oussaid, L. (2016). Ontological Interaction Modeling and Semantic Rule-based Reasoning for User Interface Adaptation. In *Proceedings of the 12th International Conference on Web Information Systems and Technologies - Volume 1: SRIS, (WEBIST 2016)* (pp. 347–354). Rome, Italy: SciTePress. INSTICC. doi:10.5220/0005854303470354.
- Li, Y., Rossmy, B. & Hußmann, H. (2020). Tangible Interaction with Light: A Review. Multimodal Technologies and Interaction, 4(4), 72. doi:10.3390/mti4040072.
- Maalej, A. & Kallel, I. (2020). Does Keystroke Dynamics tell us about Emotions? A Systematic Literature Review and Dataset Construction. In 2020 16th International Conference on Intelligent Environments (IE) (pp. 60–67). IEEE. doi:10.1109/IE49459.2020.9155004.
- Mahieu, C., Ongenae, F., Backere], F.D., Bonte, P., Turck], F.D. & Simoens, P. (2019). Semantics-based platform for contextaware and personalized robot interaction in the internet of robotic things. *Journal of Systems and Software*, 149, 138–157. doi:10.1016/j.jss.2018.11.022.
- Martín, A., Rossi, G., Cechich, A. & Gordillo, S. (2010). Engineering Accessible Web Applications. An Aspect-Oriented Approach. *World Wide Web*, *13*(4), 419–440. doi:10.1007/s11280-010-0091-3.
- Meng, Z. (2018). Architecture Support for Context-aware Adaptation of Rich Sensing Smartphone Applications. KSII Transactions on Internet and Information Systems, 12(1), 248–268. doi:10.3837/tiis.2018.01.012.
- Messaoudi, R., Mtibaa, A., Vacavant, A., Gargouri, F. & Jaziri, F. (2020). Ontologies for Liver Diseases Representation: A Systematic Literature Review. *Journal of Digital Imaging*, 33(3), 563–573. doi:10.1007/s10278-019-00303-2.
- Mezhoudi, N. & Vanderdonckt, J. (2015). A user's feedback ontology for context-aware interaction. In Proceedings of the 2015 2nd World Symposium on Web Applications and Networking (WSWAN) (pp. 1–7). Sousse, Tunisia: IEEE. doi:10.1109/WSWAN.2015.7210331.
- Myrgioti, E., Bassiliades, N. & Miliou, A. (2013). Bridging the HASM: An OWL ontology for modeling the information pathways in haptic interfaces software. *Expert Systems with Applications*, 40(4), 1358–1371. doi:10.1016/j.eswa.2012.08.053.
- Negru, S. & Buraga, S. (2013). A Knowledge-Based Approach to the User-Centered Design Process. In A. Fred, J.L.G. Dietz, K. Liu and J. Filipe (Eds.), Proceedings of the Knowledge Discovery, Knowledge Engineering and Knowledge Management. IC3K 2012. Communications in Computer and Information Science, vol 415. (pp. 165–178). Berlin, Heidelberg: Springer Berlin Heidelberg. doi:10.1007/978-3-642-54105-6\_11.
- Negru, S. & Buraga, S.C. (2012). Persona Modeling Process: From Microdata-based Templates to Specific Web Ontologies... In J. Filipe and J. Dietz (Eds.), *Proceedings of the International Conference on Knowledge Engineering and Ontology Development - Volume 1: KEOD*, (IC3K 2012) (pp. 34–42). Barcelona, Spain: SciTePress. doi:10.5220/0004127000340042.
- Norman, D.A. (1986). Cognitive engineering. In D.A. Norman and S.W. Draper (Eds.), User Centered System Design: New Perspectives on Human-Computer Interaction (1st ed., pp. 31–61). Hillsdale, NJ, USA: L. Erlbaum Associates Inc. Chapter 3.
- Petersen, K., Vakkalanka, S. & Kuzniarz, L. (2015). Guidelines for conducting systematic mapping studies in software engineering: An update. *Information and Software Technology*, 64, 1–18. doi:10.1016/j.infsof.2015.03.007.
- Preece, J., Sharp, H. & Rogers, Y. (2015). Interaction Design: Beyond Human-Computer Interaction (4th ed., p.584). John Wiley & Sons.
- Quirino, G.K.d.S., Barcellos, M.P. & Falbo, R.d.A. (2018). Visual notations for software pattern languages. In *Proceedings of the XXXII Brazilian Symposium on Software Engineering SBES '18* (pp. 72–81). New York, New York, USA: ACM Press. doi:10.1145/3266237.3266266.
- Real, G.d.A.C.V., Nardi, J., Barcellos, M.P. & Falbo, R.d.A. (2019). Semantic Enterprise Application Integration at Service Layer: A Systematic Mapping. In *Proceedings of the XXII Ibero-American Conference on Software Engineering (CIbSE* 2019) (p.14).
- Robal, T., Marenkov, J. & Kalja, A. (2017). Ontology Design for Automatic Evaluation of Web User Interface Usability. In Proceedings of the 2017 Portland International Conference on Management of Engineering and Technology (PICMET) (pp. 1–8). Portland, OR, USA: IEEE. doi:10.23919/PICMET.2017.8125425.
- Rodrigues, C.M.d.O., Bezerra, C., Freitas, F. & Oliveira, I. (2020). Handling Crimes of Omission by reconciling a criminal core ontology with UFO. *Applied Ontology*, 15, 7–39. doi:10.3233/AO-200223.

- Romero Mariño, B., Rodríguez Fórtiz, M.J., Hurtado, M. & M. Haddad, H. (2018). Accessibility and Activity-Centered Design for ICT Users: ACCESIBILITIC Ontology. *IEEE Access*, 6, 60655–60665. doi:10.1109/ACCESS.2018.2875869.
- Salamon, J.S., Reginato, C.C. & Barcellos, M.P. (2019). Ontology Integration Approaches: A Systematic Mapping. In Proceedings of the Brazilian Seminar on Ontologies (ONTOBRAS), CEUR Workshop (p.11).
- Salma, G. & Marouane, e.M. (2016). Operating an application for modeling persona by using ontologies. *Journal of Theoretical & Applied Information Technology*, 88(1), 57–64. https://www.scopus.com/inward/record.uri?eid=2-s2. 0-84973549972{&}partnerID=40{&}md5=822eb166a69c5fe36c68dd8e66ef6913.
- Salma, G., Eddine, M.E.K.K. & Sabin, M.C.B. (2012). Use of ontologies in modeling persona. In Proceedings of the 2012 IEEE International Conference on Complex Systems (ICCS) (pp. 1–9). Agadir, Morocco: IEEE. doi:10.1109/ICoCS.2012.6458521.
- Satyanarayanan, M. (2001). Pervasive computing: vision and challenges. *IEEE Personal Communications*, 8(4), 10–17. doi:10.1109/98.943998.
- Sauer, J., Sonderegger, A. & Schmutz, S. (2020). Usability, user experience and accessibility: towards an integrative model. *Ergonomics*, 63(10), 1207–1220. doi:10.1080/00140139.2020.1774080.
- Scherp, A., Saathoff, C., Franz, T. & Staab, S. (2011). Designing core ontologies. Applied Ontology, 6(3), 177–221. doi:10.3233/AO-2011-0096.
- Schiaffino, S. & Amandi, A. (2009). Intelligent User Profiling. In M. Bramer (Ed.), Artificial Intelligence An International Perspective: An International Perspective (pp. 193–216). Berlin, Heidelberg: Springer Berlin Heidelberg. doi:10.1007/978-3-642-03226-4\_11.
- Shahzad, S.K., Granitzer, M. & Helic, D. (2011). Ontological model driven GUI development: User interface ontology approach. In Proceedings of the 6th International Conference on Computer Sciences and Convergence Information Technology, ICCIT 2011 (pp. 214–218). Seogwipo, South Korea: IEEE.
- Silva, T.R., Winckler, M. & Trætteberg, H. (2019). Ensuring the Consistency Between User Requirements and GUI Prototypes: A Behavior-Based Automated Approach. In D. Lamas, F. Loizides, L. Nacke, H. Petrie, M. Winckler and P. Zaphiris (Eds.), Proceedings of the Human-Computer Interaction – INTERACT 2019. INTERACT 2019. Lecture Notes in Computer Science, vol 11746 (pp. 644–665). Cham: Springer International Publishing. doi:10.1007/978-3-030-29381-9\_39.
- Sokolowski, J. & Walczak, K. (2018). Semantic Modelling of User Interactions in Virtual Reality Environments. In L.M. Camarinha-Matos, K.O. Adu-Kankam and M. Julashokri (Eds.), Proceedings of the Technological Innovation for Resilient Systems. DoCEIS 2018. IFIP Advances in Information and Communication Technology, vol 521 (pp. 18–27). Cham: Springer International Publishing. doi:10.1007/978-3-319-78574-5\_2.
- Sousa, A., Couto, T., Agra, C. & Alencar, F. (2016). Use of Ontologies in Embedded Systems: A Systematic Mapping. In A.V.M.R.J.G.M. Paulk M. Brito M.A. (Ed.), Proceedings of the 10th International Conference on the Quality of Information and Communications Technology (QUATIC) (pp. 1–8). IEEE. doi:10.1109/QUATIC.2016.011.
- Storrs, G. (1994). A conceptualisation of multiparty interaction. *Interacting with Computers*, 6(2), 173–189. doi:10.1016/0953-5438(94)90023-X.
- Studer, R., Benjamins, V.R. & Fensel, D. (1998). Knowledge engineering: Principles and methods. Data & Knowledge Engineering. doi:10.1016/S0169-023X(97)00056-6.
- Suàrez, P.R., Jùnior, B.L. & de Barros, M.A. (2004). Applying knowledge management in UI design process. In P. Slavik and P. Palanque (Eds.), *Proceedings of the 3rd annual conference on Task models and diagrams - TAMODIA '04* (pp. 113–120). NY, NY, USA: ACM Press. doi:10.1145/1045446.1045468.
- Suárez-Figueroa, M.C., Gómez-Pérez, A., Motta, E. & Gangemi, A. (2012). Ontology Engineering in a Networked World (p.XII, 444). Springer-Verlag Berlin Heidelberg. doi:10.1007/978-3-642-24794-1.
- Surve, A.R. & Ghorpade, V.R. (2017). Pervasive Context-Aware Computing Survey of Context-aware ubiquitous middleware systems. *International Journal of Engineering Research and Technology*, *10*(1), 974–3154.
- Sutcliffe, A.G. (2014). Requirements Engineering from an HCI Perspective. In M. Soegaard and R.F. Dam (Eds.), *The Encyclopedia of Human-Computer Interaction* (2nd ed., pp. 707–760). Aarhus, Denmark: The Interaction Design Foundation. Chapter 13.
- Tapia-Leon, M., Rivera, A.C., Chicaiza, J. & Lujan-Mora, S. (2018). Application of ontologies in higher education: A systematic mapping study. In *Proceedings of the IEEE Global Engineering Education Conference (EDUCON)* (Vol. 2018-April, pp. 1344–1353). IEEE. doi:10.1109/EDUCON.2018.8363385.
- Tarhan, A. & Giray, G. (2017). On the Use of Ontologies in Software Process Assessment. In Proceedings of the 21st International Conference on Evaluation and Assessment in Software Engineering. EASE'17 (pp. 2–11). New York, NY, USA: ACM. doi:10.1145/3084226.3084261.
- Tourwé, T., Tsipokova, E., González-Deleito, N. & Hristoskova, A. (2011). Ontology-driven elicitation of multimodal user interface design recommendations. In P. De Causmaecker, J. Maervoet, T. Messelis, K. Verbeeck and T. Vermeulen (Eds.), *Proceedings of the BNAIC : Belgian/Netherlands Artificial Intelligence Conference* (p.8). Ghent, Belgium. doi:1854/LU-2012408.

- Victoria, I., Antoanela, N. & Cicortas, A. (2008). Deriving ontologies using multi-agent systems. WSEAS Transactions on Computers, 7.
- Wohlin, C., Runeson, P., Höst, M., Ohlsson, M.C., Regnell, B. & Wesslén, A. (2012). Systematic Literature Reviews. In Experimentation in Software Engineering (pp. 45–54). Berlin, Heidelberg: Springer Berlin Heidelberg. doi:10.1007/978-3-642-29044-2\_4.
- Zamzami, E.M., Budiardjo, E.K. & Suhartanto, H. (2013). Requirements recovery using ontology model for capturing end-toend interaction of proven application software. *International Journal of Software Engineering and its Applications*, 7(6), 425–434. doi:10.14257/ijseia.2013.7.6.36.
- Zárate, M., Braun, G., Fillottrani, P., Delrieux, C. & Lewis, M. (2020). BiGe-Onto: An ontology-based system for managing biodiversity and biogeography data. *Applied Ontology, Preprint*, 1–27. doi:10.3233/AO-200228.